Author’s response to reviews

Title: Recursive neural networks in hospital bed occupancy forecasting

Authors:

Ekaterina Kutafina (ekutafina@mi.rwth-aachen.de)
Istvan Bechtold (istvan.bechtold@st-marienkrankenhaus.de)
Klaus Kabino (kkabino@mi.rwth-aachen.de)
Stephan Jonas (sjonas@mi.rwth-aachen.de)

Version: 2 Date: 13 Feb 2019

Author’s response to reviews:

Dear Editors,

Please find below the list of changes.

1. Figures are provided as separate files.

2. Titles/legends are provided under a separate heading of 'Figure Legends' after the References.

3. We declare, that no administrative permissions were required to access the raw data.

4. The list of contributions was changed to better reflect the actual workload.

5. The highlights were removed.

We hope that the current version of the manuscript will fulfill the journal’s standard.

Best regards,

Ekaterina Kutafina